Author's response to reviews

Title: Multifocal invasive ductal breast cancer with osteoclast-like giant cells

Authors:

Georg Richter (richter@pathologie-hannover.de)
Christoph Uleer (christoph.uleer@t-online.de)
Thomas Noesselt (noesselt@kreiskrankenhaus-hameln.de)

Version: 7 Date: 22 October 2010

Author's response to reviews: see over
Cover letter:

1. Please revise the consent statement to confirm that informed written consent was received for publication of the manuscript and figures:

The consent statement has been revised, it now reads as follows:

Written informed consent was obtained from the patient for publication of this case report and accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.

2. Please include a section header/heading for the Consent section:

A section header/heading for the Consent section has been included.

3. Please include a section header/heading for the Competing Interest section:

A section header/heading for the Competing Interest section has been included.

4. Please include a section header/heading for the Authors’ Contribution section:

A section header/heading for the Authors’ Contribution section has been included.